Author's response to reviews

Title: Dealing with Heterogeneity of Treatment Effects: Is the Literature Up to the Challenge? A Systematic Review

Authors:

Nicole B Gabler (nrbloser@ucdavis.edu)
Naihua Duan (Naihua.Duan@Columbia.edu)
Diana Liao (DLiao@mednet.ucla.edu)
Joann G Elmore (jelmore@u.washington.edu)
Theodore Ganiats (tganiats@ucsd.edu)
Richard L Kravitz (rlkravitz@ucdavis.edu)

Version: 2 Date: 10 October 2008

Author's response to reviews:

October 10, 2008

Doug Altman, PhD
Curt Furberg, MD, PhD
Jeremy Grimshaw, MBChB, PhD, FRCGP
Peter Rothwell, MD, PhD, FRCP
Editors-in-Chief, Trials

Dear Drs. Altman, Furberg, Grimshaw, and Rothwell:

We are submitting a revised manuscript to correct two oversights in our original submission, dated 3 October 2008. The first revision is the addition of figure legends. The second revision is the reformatting of the abstract to adhere to the structure guidelines (background, methods, results, and conclusion). All other aspects of the manuscript remain as they were in the 3 October 2008 submission.

We appreciate your consideration.

Sincerely,

Nicole Bloser Gabler, MPH, MHA
PhD Candidate, Graduate Group in Epidemiology
University of California, Davis
Center for Healthcare Policy and Research
2103 Stockton, Blvd., Grange Building, Suite 2224
Sacramento, CA 95817